

Author's response to reviews

Title: Cilostazol Improves Endothelial Function in Acute Cerebral Ischemia Patients: A Doubleblind Placebo Controlled Trial with Flow-mediated Dilation Technique

Authors:

Seong-Joon Lee (editisan@gmail.com)

Jin Soo Lee (jinsoo22@gmail.com)

Mun Hee Choi (choimoonhee09@gmail.com)

Sung Eun Lee (plumpboy@hanmail.net)

Dong Hoon Shin (sphincter@naver.com)

Ji Man Hong (dacda@hanmail.net)

Version: 1 Date: 16 Aug 2017

Author's response to reviews:

The authors' response letter has been included as a supplementary file.